Author's response to reviews

Title: Keep Cool: Effectiveness of a web-based education program to improve vaccine storage conditions in primary care: study protocol for a randomized controlled trial

Authors:

Anika Thielmann (anika.thielmann@uk-essen.de)
Anja Viehmann (anja.viehmann@uk-essen.de)
Birgitta M Weltermann (birgitta.weltermann@uk-essen.de)

Version: 2  Date: 11 March 2015

Author's response to reviews:

Dear Mrs. McGavin,

1) We included a statement in the methods section: “All participants will sign an informed consent form.”

2) We specified the funding information: “Acknowledgments: A grant for the study is being submitted. Funding for the temperature data loggers used in the study was received from the Stiftung Universitätsmedizin Essen, Germany. Funding for the educational software and its adjustments was granted by the Kulturstiftung Essen, Germany. The sources of funding do not influence the study design, analyses, interpretation of results or the decision on publication.”

We received full ethical approval with no conditions and have obtained some external funding. We have mailed you the requested documentations.

Please do not hesitate to contact us if any further questions arise.

Sincerely yours,

Birgitta Weltermann